Author’s response to reviews

Title: Impact of antiretroviral treatment on height evolution of HIV infected children

Authors:

PATRINEE TRAISATHIT (patrinee@gmail.com)
Saïk Urien (saik.urien@cch.aphp.fr)
Sophie Le Coeur (sophie.lecoeur@phpt.org)
Sakulrat Srirajana (sakulrats@yahoo.com)
Noppadon Akarathum (nakkaratham@gmail.com)
Suparat Kanjanavanit (skanjanavanit@gmail.com)
Chaiwat Ngampiyaskul (chaiwat008@gmail.com)
Sawitree Krikajornkitti (yuiwitree@gmail.com)
Nicole Ngo-Giang-Huong (Nicole.Ngo-Giang-Huong@phpt.org)
Marc Lallemant (marclallemant@gmail.com)
Gonzague Jourdain (gonzague.jourdain@ird.fr; gonzague.jourdain@phpt.org)

Version: 2 Date: 04 Aug 2019

Author’s response to reviews:

Amelia De Salis, BSc
Editor, BMC Pediatrics

Dear Ms Amelia De Salis,

Following your request, we are submitting a new version our manuscript entitled “Impact of antiretroviral treatment on height evolution of HIV infected children”.

We have addressed your comments:
- regarding the 'Funding', we have specified: ‘The funding bodies did not have any role in study design, data collection, analysis, and interpretation of data, decision to publish, or preparation of the manuscript.’

- the authors’ full names have been replaced with their initials in the Authors’ Contributions section

- we have added a ‘Consent for publication’ section and specified ‘Not Applicable’

- we have removed the response to the reviewers from the submission’s supplementary files.

We have uploaded the new version of the manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors.

I also made some minor edits in the Acknowledgements section.

Thank you for your help,

Sincerely,

Gonzague Jourdain